Author’s response to reviews

Title: The Study Protocol of EMPOWER Participatory Action Research (EMPOWER - PAR): a pragmatic cluster randomised controlled trial of multifaceted chronic disease management strategies to improve diabetes and hypertension outcomes in primary care

Authors:

Anis S Ramli (anis014@salam.uitm.edu.my)
Sharmila Lakshmanan (sharmilamkl@crc.gov.my)
Jamaiyah Haniff (jamaiyah@crc.gov.my)
Sharmini Selvarajah (s.selvarajah@um.edu.my)
Seng F Tong (tsf@ppukm.ukm.edu.my)
Adam Bujang (adam@crc.gov.my)
Suraya Abdul-Razak (suraya617@salam.uitm.edu.my)
Asrul A Shafie (aakmal@usm.my)
Verna KM Lee (verna.lee69@gmail.com)
Thuhairah H Abdul-Rahman (thuhairah@salam.uitm.edu.my)
Maryam H Daud (maryamhannahdaud@yahoo.com)
Kien K Ng (drkkgng@hotmail.com)
Farnaza Ariffin (farnaza@salam.uitm.edu.my)
Hasidah Abdul-Hamid (hasidah@salam.uitm.edu.my)
Md-Yasin Mazapuspavina (puspavina@yahoo.com.my)
Nafiza Mat-Nasir (dmnfiza220@yahoo.com)
Maizatullifah Miskan (maizatullifah@salam.uitm.edu.my)
Jaya P Stanley-Ponniah (jaya@crc.gov.my)
Mastura Ismail (drami68@yahoo.com)
Chun W Chan (chunwai_chan@imu.edu.my)
Yong R Abdul-Rahman (yongrafidah@gmail.com)
Boon H Chew (chewbh@medic.upm.edu.my)
Wilson HH Low (wilson.low@azmi-burhani.com)

Version: 5 Date: 20 August 2014

Author’s response to reviews:

Dear Executive Editor,

We noted minor mistakes in the revised version. Another copy edited version is therefore being submitted.